Author's response to reviews

Title: Differences in Risk Factors for Children With Special Health Care Needs (CSHCN) Receiving Needed Speciality Care by Socioeconomic Status

Authors:

Kristine A Lykens (klykens@hsc.unt.edu)
Kimberly G Fulda (kfulda@hsc.unt.edu)
Sejong Bae (sbae@hsc.unt.edu)
Karan P Singh (ksingh@hsc.unt.edu)

Version: 12 Date: 11 July 2009

Author's response to reviews:

July 10, 2009

Dear Editors:

As requested in your email of July 8, 2009 I am resubmitting manuscript 3173576532175765 with the text frames removed from each paragraph and the renumbering of attachments.

Thank you for the opportunity to make these correction.

Kris Lykens